Author's response to reviews

Title: Screening for inter-hospital differences in cesarean section rates in low-risk deliveries using administrative data. An initiative to improve the quality of care

Authors:

Willem Aelvoet (willem.aelvoet@health.fgov.be)
Francis Windey (francis.windey@health.fgov.be)
Geert Molenberghs (Geert.Molenberghs@uhasselt.be)
Hans Verstraelen (hans.verstraelen@Ugent.be)
Patrick Van Reempts (Patrick.Van.Reempts@uza.be)
Jean-Michel Foidart (jmfoidart@ulg.ac.be)

Version: 8 Date: 26 November 2007

Author's response to reviews:

Dear editor,

We included an abstract, structured into the 4 required sections, as well as a Title page.